Author's response to reviews

Title: Randomised controlled trial of a psychiatric consultation model for treatment of common mental disorder in the occupational health setting. Study Protocol.

Authors:

Christina M Van der Feltz-Cornelis (cfeltz@trimbos.nl)
Jolanda Meeuwissen (jmeeuwissen@trimbos.nl)
Fransina J De Jong (sjong@trimbos.nl)
Rob Hoedeman (rob.hoedeman@arboned.nl)
Iman Elfeddali (ielfeddali@trimbos.nl)

Version: Date: 21 January 2007

Author’s response to reviews:

Dr Annabel Phillips January 21th, 2007
Senior Assistant Editor
BMC-series Journals
BioMed Central
Tel: +44 (0)20 7631 9921
Facsimile: +44 (0)20 7631 9923
e-mail: editorial@biomedcentral.com
Web: http://www.biomedcentral.com/

Christina M van der Feltz-Cornelis
Psychiatrist-epidemiologist, Head of Research Program:
Diagnosis and Treatment of common mental disorders
Board of the Depression Initiative Program
Representative of the Netherlands in the EAAD
Trimbos-Instituut / Netherlands institute of Mental Health and Addiction
PO Box 725
3500 AS Utrecht
The Netherlands
Phone +31.30.2971126
Fax +31.30.2971111
email cfeltz@trimbos.nl
Senior researcher Psychiatry Dept.
Free Medical University Institute of Extramural Research,
Common mental disorders program

Dear Mrs. Philips,

Hereby I submit to you the final version of the article after it was corrected by a certified translator/native speaker. It was also formatted according to the format of your journal.
I already did send you two extra documents from the funding institution that mention the approval and the amount of funding in euro's, as you requested.
I do hope that this is sufficient information for you and that you can accept the article for publishing now. However, if you would have any remaining questions, please let me know.

Sincerely,

Christina M van der Feltz-Cornelis